Author’s response to reviews

Title: Sub-hypnotic dose of propofol as antiemetic prophylaxis attenuates intrathecal morphine-induced postoperative nausea and vomiting, and pruritus in parturient undergoing cesarean section — a randomized control trial.

Authors:

Sylvanus Kampo (sylvanuskampo@yahoo.com)
Alfred Afful (alfredo2k2@yahoo.com)
Shiraj Mohammed (shirajm90@gmail.com)
Michael Ntim (archangelmntim@yahoo.co.uk)
Alexis Buunaaaim (abuunaaaim@yahoo.co.uk)
Thomas Anabah (dranubah@gmail.com)

Version: 1 Date: 04 Apr 2019

Author’s response to reviews:

1. The TRN has been provided as suggested by the editor
2. The manuscript reporting adheres to CONSORT guidelines. The CONSORT checklist has been submitted.
3. Some authors institutional email addresses has been provided
4. The changes have been done per the editor's suggestion
5. We did the changes by using the Vancouver style of referencing as the editor suggested
5. We submitted the manuscript without track changes or highlight.